Author's response to reviews

Title: The minimal clinically important difference of six-minute walk in Asian older adults

Authors:

Boon Chong Kwok (kwokboonchong@gmail.com)
Yong Hao Pua (puayonghao@gmail.com)
Kaysar Mamun (kaysar.mamun@sgh.com.sg)
Wai Pong Wong (waipong.wong@singaporetch.edu.sg)

Version: 2 Date: 14 September 2012

Author's response to reviews:

Dear editor(s) of BMC Geriatrics,

I have included the request of "ethics" and "informed consent" into the manuscript under Methods - Participants section. Quoted here is the included text: "This study has been approved by the SingHealth Centralised Institutional Review Board D (Reference: 2010/177/D) in accordance with the International Conference on Harmonisation /Singapore Guidelines for Good Clinical Practices. Written informed consent is obtained from the participant prior to the study participation."

Thank you for the kind advice.

With regards,

Mr Kwok